

Citation: Kayem ND, Okogbenin S, Okoeguale J, Momoh M, Njoku A, Eifediyi R, et al. (2023) Seroepidemiology of Lassa virus in pregnant women in Southern Nigeria: A prospective hospital-based cohort study. PLoS Negl Trop Dis 17(5): e0011354. https://doi.org/10.1371/journal. pntd.0011354

Editor: Michael R. Holbrook, NIAID Integrated Research Facility, UNITED STATES

Received: September 21, 2022

Accepted: May 6, 2023

Published: May 22, 2023

Copyright: © 2023 Kayem et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because data cannot be deidentified at the granular level of analyses performed and contains sensitive information. Data are available by request from Irrua Specialist Teaching Hospital Research and Ethics Committee (KM 87 Benin Auchi Rd, Irrua, Edo State, Nigeria; isth.rec.2015@gmail.com) for researchers who meet the criteria for access to confidential data and obtain ethical clearance from Irrua Specialist Teaching Hospital Research Ethics Committee. RESEARCH ARTICLE

Seroepidemiology of Lassa virus in pregnant women in Southern Nigeria: A prospective hospital-based cohort study

Nzelle Delphine Kayem^{1‡}, Sylvanus Okogbenin^{2‡}, Joseph Okoeguale², Mojeed Momoh², Antonia Njoku², Reuben Eifediyi², Xavier Enodiana², Hilary Ngwu², Wilfred Irhiogbe², Yemisi Ighodalo³, Thomas Olokor³, George Odigie³, Lyndsey Castle¹, Sophie Duraffour⁴, Lisa Oestereich⁴, Prabin Dahal¹, Proochista Ariana¹, Stephan Gunther⁴, Peter Horby¹

1 Nuffield Department of Medicine, University of Oxford, Oxford, United Kingdom, 2 Department of Obstetrics and Gynaecology, Irrua Specialist Teaching Hospital, Irrua, Nigeria, 3 Institute of Lassa fever Research and Control, Irrua Specialist Teaching Hospital, Irrua, Nigeria, 4 Bernhard-Nocht-Institute for Tropical Medicine, Hamburg, Germany

‡ These authors share first authorship on this work.
* delphinekayem@gmail.com

Abstract

Background

There is limited epidemiological evidence on Lassa fever in pregnant women with acute gaps on prevalence, infection incidence, and risk factors. Such evidence would facilitate the design of therapeutic and vaccine trials and the design of control programs. Our study sought to address some of these gaps by estimating the seroprevalence and seroconversion risk of Lassa fever in pregnant women.

Methodology/principal findings

We conducted a prospective hospital-based cohort between February and December 2019 in Edo State, Southern Nigeria, enrolling pregnant women at antenatal clinic and following them up at delivery. Samples were evaluated for IgG antibodies against Lassa virus. The study demonstrates a seroprevalence of Lassa IgG antibodies of 49.6% and a seroconversion risk of 20.8%. Seropositivity was strongly correlated with rodent exposure around homes with an attributable risk proportion of 35%. Seroreversion was also seen with a seroreversion risk of 13.4%.

Conclusions/significance

Our study suggests that 50% of pregnant women were at risk of Lassa infection and that 35.0% of infections might be preventable by avoiding rodent exposure and conditions which facilitate infestation and the risk of human-rodent contact. While the evidence on rodent exposure is subjective and further studies are needed to provide a better understanding of the avenues of human-rodent interaction; public health measures to decrease the risk of rodent infestation and the risk of spill over events may be beneficial. With an estimated

Funding: PH and NDK received a grant from the Global Challenges Research Fund (GCRF) through an Internal University of Oxford GCRF Research England Fund (Ref: 0006059, code: KCD00140-B903.01) https://globalresearch.admin.ox.ac.uk/ gcrf-oda/gcrf. NDK was also funded as a Commonwealth Scholar by the UK Department for International Development, https://cscuk.fcdo.gov. uk/scholarships/commonwealth-phdscholarshipsfor-least-developed-countries-andfragile-states/. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.ection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

seroconversion risk of 20.8%, our study suggests an appreciable risk of contracting Lassa fever during pregnancy and while most of these seroconversions may not be new infections, given the high risk of adverse outcomes in pregnancy, it supports the need for preventative and therapeutic options against Lassa fever in pregnancy. The occurrence of seroreversion in our study suggests that the prevalence obtained in this, and other cohorts may be an underestimate of the actual proportion of women of childbearing age who present at pregnancy with prior LASV exposure. Additionally, the occurrence of both seroconversion and seroreversion in this cohort suggests that these parameters would need to be considered for the development of Lassa vaccine efficacy, effectiveness, and utility models.

Author summary

Lassa virus is a viral haemorrhagic fever endemic to West Africa and yet to date there is limited information on its epidemiology in pregnancy. Information obtained from epidemiology studies is important because such information facilitates the design and costing of disease control programs and the design of clinical trials. Our study evaluated the epidemiology of Lassa fever in pregnant women specifically we looked at the prevalence, incidence, and risk factors for infection in pregnant women in South Nigeria. We found that 49.6% of pregnant women had antibodies to Lassa virus suggesting about 50% of pregnant women were at risk of Lassa infection. Additionally, while, the evidence was not conclusive, we found an appreciable risk of contracting Lassa fever during pregnancy. Given that, Lassa fever has high case fatality in pregnancy, our data supports the need for vaccines and drugs against Lassa fever. Our study also suggests that 35.0% of infections might be preventable by avoiding rodent exposure and conditions which facilitate rodent infestation and the risk of human-rodent contact. This suggests that public health measures such as proper sanitation and hygiene, improved housing and health education may be beneficial in reducing the risk of infection.

Introduction

Lassa fever is a zoonotic infection endemic to the West African subregion where cases are reported throughout the year, with peak incidences between November and May in endemic countries. [1–3] Modelling studies in Sub-Saharan Africa suggest that about 37.7 million people in 14 West African countries are at risk of infection [4] with an estimated 5,000 to 60,000 deaths annually. [5, 6]

Pregnant women and their foetuses are particularly prone to adverse outcomes from Lassa virus infection. A recent review suggest that Lassa fever poses an increased risk of mortality in pregnant women when compared to non-pregnant women (three times greater odds). [7] The review also showed high rates of foetal (61.5%) and neonatal losses (30.2%) suggesting that pregnant women, their foetuses and new-borns are particularly prone to adverse outcomes from Lassa fever. [7] Additionally, reports from West Africa suggest that Lassa fever is likely a major contributor to maternal mortality in endemic areas. [8, 9] Given that Lassa virus infection in pregnancy poses a significant risk of infection and adverse outcomes to the pregnant woman, and her foetus; pregnant women should be considered a 'special group' for interventions to prevent Lassa virus infection and disease. [7]

There are currently no licensed antivirals for the management of Lassa fever. [10] Ribavirin is used off-label for the management of Lassa fever. [10] However, the evidence justifying the use of ribavirin is poor, [11] and underscores the need for improved care and trials to obtain safety and efficacy data for therapeutics, [12] particularly in pregnancy. There are currently no vaccines for Lassa fever, however, amongst the viral haemorrhagic fevers, Lassa fever has one of the most advanced vaccine development platforms with six vaccines funded by The Coalition for Epidemic Preparedness Innovations (CEPI) which supports development of vaccines for emerging pathogens. [13] Most vaccine candidates are in preclinical trials, but there are currently three vaccines in Phase I human trials, Inovio's DNA vaccine completed recruitment in 2022 for a Phase IB trial in Ghana; [14, 15] while IAVI's vaccine and Emergent BioSolutions's vaccine began Phase I First in human clinical trials in Liberia and Ghana respectively. [15] This underscores the need for further studies particularly clinical trials.

Inclusion of pregnant women in therapeutic trials would improve our understanding of the effects of various treatments in pregnancy. Similarly, inclusion of pregnant women in future vaccine trials would facilitate evaluation of vaccine efficacy and effectiveness in the context of pregnancy and provide evidence for vaccine program implementation, ensuring that pregnant women are not excluded from the benefits of vaccination. [12] However, both vaccine and therapeutic studies require evidence on the disease epidemiology.

Epidemiological evidence to support the design of vaccine or therapeutic trials is significantly limited with critical gaps in data on prevalence, infection incidence, risk groups, disease presentation, and antibody kinetics, and these gaps are even more acute for pregnant women. [7, 16] Evidence on prevalence and incidence facilitates the assessment of disease burden and the identification of risk factors for infection, which in turn support the development of treatment and control strategies and the planning and allocation of resources.

In the general population, over 80% of Lassa virus infections are undiagnosed or unreported, [5, 17] as such, the rates of Lassa virus infection (incidence estimates) inferred from clinically apparent cases alone would be a substantial underestimate. The best method for estimation of the Lassa fever burden would be through serological testing; that is, the prevalence of antibodies in the population of interest using assays that are highly sensitive and specific. [18] Data from West Africa have shown a high prevalence of Lassa virus antibodies (LASV IgG) in the general population with wide variation between and within countries, ranging from 8–52% in Sierra Leone, [5, 6] 21–58% in Nigeria, [19, 20] 4–55% in Guinea, [2, 5, 21] to 3–14% in Liberia. [22, 23] Annual seroconversion rates in the general population are estimated to range from 4.2–8.4% in Mali [24] to 5–22% in Sierra Leone. [6] Among pregnant women, one estimate of prevalence, obtained from a single study in Nigeria over a nine-year period, was 0.87% (44/5048). [9] This period prevalence was based solely on active cases or symptomatic Lassa fever pregnant women, and as such, is possibly a considerable underestimation of Lassa prevalence in pregnancy. It is likely that, similar to the general population, most Lassa virus infections in pregnancy are undiagnosed.

To the best of our knowledge, no previous studies have estimated Lassa seroprevalence, seroconversion or seroincidence in pregnancy. This study sets out to estimate the prevalence of prior LASV exposure (seroprevalence) and the incidence of new Lassa virus infections (seroconversion) in pregnant women and evaluate risk factors associated with seroconversion and seroprevalence for a better understanding of the epidemiology of Lassa fever in preparation for future Lassa vaccine and therapeutic trials.

Methods

Ethics statement

Ethical approval was obtained from the Oxford Tropical Research Ethics Committee (OxTREC) of the University of Oxford (OxTREC reference No.: 49–18), and the Irrua Specialist Teaching Hospital Research Ethics Committee (ISTH REC), Edo State, Nigeria (HREC Approval No.: NHREC/29/03/2017). Written informed consent was obtained from all participants.

Study endpoints

To the best of our knowledge, no previous studies have estimated seroprevalence or seroconversion in pregnant women alone and as such we set it up as an exploratory study with no predefined hypothesis. <u>Table 1</u> summarises the objectives of our study and the expected outcomes.

Definition of endpoints

Further details on Lassa antibodies and the ELISA used for this study are found in the section on sample collection and laboratory analysis.

Seropositivity was defined based on cut-offs for LASV IgG antibody levels for the ELISA, as such LASV seropositivity was defined as samples with an antibody index value of 1.1 or above (≥ 1.1) .

Seronegativity was defined as antibody index values of 0.9 or below (\leq 0.9). Susceptible pregnant women were defined as the proportion of pregnant women who were negative for IgG antibodies.

Seroconversion was defined as a participant who was serone gative for LASV IgG at baseline (index value ≤ 0.9), became seropositive at delivery (IgG index value ≥ 1.1), was in the study for at least 30 days, (that is the interval between the first maternal sample and the second maternal sample was ≥ 30 days) and had a change in the index value of four-fold or more.

A lower limit of 30 days was chosen because studies suggest that IgG has a mean time to first detection of 25.6 ± 3 days after symptom onset. [25] A four-fold rise in the index value was used because a four-fold increase in antibody titre is conventional for an antibody change defining infection and seroconversion. [26, 27] A fold change was calculated as the ratio of antibody concentration at delivery to that at enrolment.

Seroreversion was not an initial objective and was estimated post-hoc because of its relevance for disease burden estimates and preventative strategies. No standard definition exists for LASV IgG seroreversion, thus we chose to define seroreversion as a change from positive (index value \geq 1.1) to negative (index value \leq 0.9) plus a four-fold decrease in concentration. [6]

•	•	
	Objectives	Outcome measures
Primary objectives	To estimate the proportion of the pregnant population susceptible to Lassa fever in Irrua, Nigeria	Seroprevalence of LASV IgG antibodies in the study population
	To estimate the proportion of pregnant women seroconverting between first and second blood collection.	The incidence proportion of seroconversion (seroconversion risk)
Secondary objectives	To assess factors associated with maternal Lassa virus (LASV) seropositivity in pregnant women	Association between putative risk factors and IgG seropositivity (measured by odds ratio)
Post-hoc	To estimate the proportion of women seroreverting between first and second blood collection	Seroreversion risk

Table 1. Study objectives and outcomes measures.

https://doi.org/10.1371/journal.pntd.0011354.t001

Study population and study procedures

The study was designed as a prospective hospital-based cohort, with pregnant women enrolled at the antenatal clinic and followed up at delivery to assess seroconversion rates. The inclusion criteria were pregnant women aged 15 years and above, attending antenatal clinics (ANC), and who were willing and able to give informed consent. Exclusion criteria comprised failure to give consent, withdrawal of consent, contraindication to venepuncture, absent or unavailable legally authorised representative, and patients for end-of-life care.

Participants were recruited between 12th February 2019 to 4th May 2019 from antenatal clinics (ANC) at Irrua Specialist Teaching Hospital (ISTH) and two of health centres in Edo, Southern Nigeria and followed up to delivery with the last participant delivered on December 20th, 2019. The health centres (Usugbenu primary centre and Eromosele medical centre) were randomly chosen from a list of facilities that frequently refer patients to ISTH and that were willing to participate in research studies. This period covered the known peak season for Lassa incidence in Edo state and in Nigeria which runs from December to May particularly January to March. [3, 28]

Evidence suggest that in developing countries while over 70% of pregnant women may attend ANC at least once, only 20–40% deliver in a health facility. [29, 30] Given this substantial risk of loss to follow up and coupled with the asymptomatic nature of Lassa fever; the sample size calculation was carried out based on a cross-sectional study design to detect seroprevalence at two time points (ANC and delivery). Evidence in the general population suggest that Lassa seroprevalence ranges from 21–58% in Nigeria [19, 20] taking a conservative prevalence of seropositivity of 30% and an absolute precision of 7.5% at a 95% confidence interval, the sample size was calculated at 144 (S1 File). However, demographic surveillance at Irrua Specialist Teaching Hospital (ISTH) shows that only 50–65% of pregnant women who attended ANC at ISTH deliver at ISTH. Thus, with an estimated loss to follow up of 40% between enrolment (ANC) and delivery, the sample size was increased to 240 pregnant women.

At the ANC, a member of the study team (health care workers) introduced the study to all pregnant women attending the health education sessions. A stratified sampling technique was used where every third pregnant woman was screened for eligibility. Participants who were eligible and consented were enrolled in the study (S2 File). These participants were followed up at delivery. Fig 1 shows the flow diagram for enrolment.

A structured questionnaire was verbally administered in local languages (English and Pidgin) at enrolment and within 48 hours of delivery to collect clinical and demographic information. At enrolment we collected information on age, gestational age, residence in the 6months prior to enrolment, educational level, occupation, knowledge of Lassa fever risk factors and transmission, history of rodent exposure in the 6 months prior to enrolment, history of fever in the two weeks before enrolment, history of Lassa fever and other medical conditions such as diabetes, HIV, and hypertension, history of pregnancy related complications such as pregnancy induced hypertension (S3 File). A second questionnaire was completed within 48 hours of delivery to collect clinical data on residence between ANC and delivery, mode of delivery, birth weight of the neonate, sex of neonate, gestational age at birth, Apgar score at birth, fever between ANC and delivery, medical conditions between ANC and delivery, pregnancy related complications between ANC and delivery. (S4 File).

Data were later entered and managed using REDCap (Research Electronic Data Capture) electronic data capture tools. [31] The gestational age was estimated using fundal height and obstetric ultrasound at enrolment and using the Ballard Score [32] at birth. Clinical data on birthweights, malaria, HIV, Lassa fever, diabetes, hypertension, and other medical conditions



Fig 1. Flow diagram for participant recruitment. Note: Recruitment took place from 12th February to 4th May 2019. The number of possible participants who left before they were contacted is large because the study team had to divide their time between busy work schedules and the study.

https://doi.org/10.1371/journal.pntd.0011354.g001

were based on patient responses and verified using patient-held ANC books and/or patient hospital records. During training for data collection staff were told if patients reported illness to cross check in case notes (hospital records or ANC booklet); previous illness data was not further differentiated as self-reported, or laboratory confirmed illness. Gestational age below 37 weeks was defined as prematurity, and birthweight below 2.5Kg was defined as low birth weight. Rat exposure was retrospective and self-reported, here, the term rat is used to refer to small rodents found peridomestically; participants were asked about a history of eating rodents, rodents in homes, rodent burrows around the house and killing rodents found in homes in the 6 months prior to enrolment. We did not provide any advice on Lassa control as part of this study.

Sample collection and laboratory analysis

Immune and pathophysiologic responses to Lassa virus are poorly understood. LASV-IgM rises by the second week of illness, [2, 25, 33, 34] with a mean time to first detection of 13.1 ± 0.9 days post-symptom onset. [25] Unlike in some viral infections, LASV IgM is not an appropriate marker for acute infection because LASV IgM can last over 12 months, [34] has been found in healthy controls, [34] may be falsely negative early in infection, and in endemic areas can lead to false positives due to immune responses to other common infections such as malaria. [35] As such, LASV IgM is possibly of 'little or no diagnostic value', particularly in endemic areas. [34, 35]

Studies suggest LASV IgG antibodies are produced in about the third week of illness [25] with a mean time to first detection of 25.6 ± 3 days after symptom onset. [25] LASV-IgG concentration may remain constant for over 12months. [34] The presence of LASV IgG is usually suggestive of past infection. However, LASV IgM and IgG may rise simultaneously, [35] and LASV IgG has been detected in some patients within a few days of infection, [1, 33, 36] and in

some cases, was detected earlier than IgM. [35] As such, recent infection and seroconversion may also be diagnosed by detection of LASV IgG particularly in follow-up samples. [35]

LASV has a high genetic diversity as such, there is currently no gold standard or reference assay for definitive laboratory diagnosis or serological testing of Lassa fever. [37, 38] The assay used in this study was one developed and previously used in this region showing a high specificity of 95–100%. [35]

Whole blood samples (2 - 5mL) were collected from mothers at enrolment and within 48 hours of delivery, and from the placental end of the umbilical cord immediately upon delivery. Samples were then centrifuged and aliquots (0.1–0.2mL) stored at –20°C until February 2020 when LASV IgG antibody ELISA (enzyme-linked immunosorbent assay) was performed using BLACKBOX LASV IgG ELISA (Diagnostics Development Laboratory, Bernhard-Nocht Institute for Tropical Medicine, Hamburg, Germany). [35] It is an immune-complex binding ELISA with antibody levels expressed as an index value (IV), calculated using optical density (OD) values. An index value of 1.1 or above (>1.1) was considered positive for past or current infection, 0.9 or below (\leq 0.9) was negative, and values between 0.9 and 1.1 were equivocal. Serology was performed following the manufacturer's instruction, which can be obtained freely from the European Virus Archive at https://www.european-virus-archive.com. All samples were stored on site at ISTH in Nigeria and while all contingencies were taken to ensure cold chain maintenance. In February 2020, prior to analysis, a fault in the backup generator resulted in our samples thawing, samples were refrozen but given that there is no data on the effect of freeze thaw cycles on Lassa antibodies, our findings need to be taken in light of this consideration.

Statistical analysis

Data were analysed using R statistical software version 4.0.2. [39] Descriptive data analyses were performed to summarise the main characteristics of pregnant women. Summary statistics were presented as mean, median or frequencies depending on the variable. Non-parametric tests were used unless the Shapiro-Wilks test was not significant.

Overall seroprevalence was defined as pregnant women who were seropositive at any point during the study period. The numerator was pregnant women who were LASV IgG positive either at baseline or at delivery (Index value ≥ 1.1), and the denominator was the total number of pregnant women recruited into the study (N = 240). The 95% confidence intervals (CI) for the proportion was calculated using Wilson/Brown intervals for binomial proportions. [40]

Univariable logistic regression was performed to evaluate the association of different maternal characteristics with overall seropositivity, and these were reported as crude odds ratios (OR) with their corresponding 95% CI. Given that the overall seroprevalence included a varying number of visits (enrolment ± delivery), we performed an initial analysis to identify factors associated with seropositivity after controlling for the number of visits (baseline model). Factors that were remained marginally significant in the baseline model (P < 0.25) were included in the multivariable model. [41] Given past reports on a link between rodent exposure and Lassa fever, [42, 43] we calculated an attributable risk proportion for rodent exposure (S1 File).

Seroconversion risk was calculated as the proportion of pregnant women in the study population who seroconverted amongst the susceptible pregnant women (that is those who were negative for LASV IgG at baseline). The 95% CI for the seroconversion risk was calculated using Wilson/Brown intervals for binomial proportions.[40] Maternal factors were assessed for their association with seroconversion using the Fisher's exact test, χ^2 test and Mann-Whitney U test as appropriate, and a logistic regression was performed. Factors that were significant in the univariable model (P < 0.25) were included in the multivariable model. [41] When performing a conventional regression with small sample sizes or rare events, there is a risk of quasi or complete separation of the model, and a likelihood of underestimating the true effect size; as such we used the Firth or penalised maximum likelihood regression which allows for calculation of finite and consistent estimates in the regression parameters. [44, 45]

We estimated seroreversion risk post-hoc. Seroreversion was calculated as the number of pregnant women who were LASV IgG positive (index value ≥ 1.1) at baseline but were seronegative (index value ≤ 0.9) at delivery.

For all regressions, there were no imputations for missing data; consequently, denominators vary by response. Analysis was conducted in three steps. First, we performed a univariate analysis to explore variables for inclusion in a multivariable model. A P value of <0.25 was set as cut-off, that is the "*P to remove*". [46] Next, we performed a multivariable regression where a two-sided P-value of ≤ 0.05 was considered statistically significant. However, to account for multiple comparisons and given that the study was exploratory and does not have a predefined hypothesis, the Bonferroni correction was applied. [47] To prevent loss of information and the risk of increasing type-1 error, continuous variables were not categorised in the regression models, [48, 49] thus in all regressions, age, gestational age, and parity were not categorised, when evaluating their correlation with seropositivity or seroconversion.

Collinear terms for the study were age and parity; gestational age at birth and birth weight; and fever during pregnancy and malaria during pregnancy. Malaria, age and gestational age at birth are more biologically relevant for changes in immune responses (antibody changes). [35, 50–52] As such, if both collinear terms were significant in the univariable model, then only age, malaria, or gestational age at birth were included in the multivariable model.

Results

The study results were reported following the Reporting Of SeroEpidemiologic Studies for Influenza (ROSES–I) statement, [53] which is easily adaptable to other infections.

Baseline characteristics

A total of 240 women were enrolled into the study. Most participants were recruited during the known peak incidence period of Lassa fever in Nigeria which runs from December to May with the highest peaks observed between January and March. [3].

The mean age of the study participants was 31.6 ± 5.0 years [range: 20–46 years] and pregnant women above 30 constituted 55.8% of the total study population. Most participants (79.1%) had at least one previous pregnancy (range: 2–8) with the majority in their second trimester of pregnancy (51.7%) and a median gestational age of 26 weeks [interquartile range (IQR): 18–31 weeks]. All participants had some form of formal education, 69.6% had completed secondary school and 54.2% lived in rural areas. Knowledge of Lassa fever risk factors and transmission was generally poor (87.5%). One hundred participants (41.7%) were employed in the informal sector (mainly as traders, tailors, hairdressers, and farmers). Most participants were from Esan West secondary administrative district or local government area (LGA), Fig 2. Details of the study participants' characteristics at baseline are provided in Table 2.

Lassa IgG seroprevalence

The overall seroprevalence (at baseline or delivery) during the study period was 49.6% [95% CI: 43.3–55.9%], with only 1.3% (3/240) of the seropositive participants reporting a prior history of a Lassa fever diagnosis. Serostatus was available at baseline for all participants and was available at delivery for 172 participants. The seroprevalence at enrolment was 40.4% [95% CI:





https://doi.org/10.1371/journal.pntd.0011354.g002

34.0–46.3%], and at delivery was 45.7% [95% CI: 38.4–53.1%]. This shows an increase from baseline; however, this increase was not statistically significant (McNemar's test; p = 0.052). The mean age among seropositive mothers was 31.8 ± 5.4 years [range: 20–45] and was not significantly different from that of seronegative mothers 31.3 ± 4.6 [range: 22–46] (t-test; p = 0.345). The median gestational age at recruitment was 24 weeks [IQR: 18–30] among seropositive mothers and 26 weeks [IQR: 21–32] among seronegative women (Mann–Whitney p = 0.276). The median time spent in the study was 107 days [Interquartile range (IQR): 55–154 days].

In a multivariable logistic regression, a self-reported prior history of Lassa fever was not significantly associated with seropositivity, and residence in a rural area was not significantly associated with seropositivity (OR, 1.42 [95% CI: 0.89–2.68], p = 0.126). Before accounting for multiple comparisons, a positive history of rodent exposure (OR, 2.32 [95% CI: 1.18–4.70, p = 0.015]) and HIV infection (OR, 5.16 [95% CI: 1.22–36.4, p = 0.046]) were significantly associated with LASV IgG seropositivity, Table 3. However, when accounting for multiple comparisons using the Bonferroni correction, none of the variables remained statistically significant.

Characteristics	n	%
Total	240	
Age (years)–Mean age ± SD [range]	31.57 ± 5.01 [20-46]	
≤ 25	33	13.8
26-30	73	30.4
31–35	90	37.5
36-40	37	15.4
> 40	7	2.9
Gestational age ^a (weeks)-Median [IQR]	26 [18-31]	
\leq 13	34	14.5
14–27	121	51.7
> 28	79	33.8
Lives in a rural area	130	54.2
Educational level ^b		
Primary	11	4.6
Secondary	62	25.8
Completed Secondary (Post-secondary)	167	69.6
Occupation		
Student	15	6.3
Housewife	39	16.3
Health Professional	25	10.4
Informal Sector	100	41.7
Formal Sector	61	25.4
Parity		
0	50	20.8
1-2	98	40.8
\geq 3	92	38.3
Poor knowledge of Lassa fever transmission and risk factors	210	87.5
Positive history of exposure to rodents in the last 6 months	191	79.6
Possible exposure to Lassa fever infected patients in last 6 months	14	5.8
Positive history of fever during pregnancy	14	5.8
Positive history of Lassa fever	5	2.1
Positive history of malaria during pregnancy	64	26.8
Hypertension	5	2.1
Pregnancy-induced hypertension	2	0.8
Diabetes mellitus	6	2.5
Gestational diabetes	3	1.3
HIV-infected	14	5.8

Table 2. Baseline characteristics of pregnant women.

The table describes the characteristics of study participants at baseline.

Note: DM- diabetes mellitus; HIV- human immunodeficiency virus; IQR- interquartile range; n- number of pregnant women recruited into the study; SD- standard deviation.

^a Data missing for 6 women.

^b All participants had some form of formal education.

https://doi.org/10.1371/journal.pntd.0011354.t002

The attributable risk proportion due to rodent exposure was estimated to be 35.03%. Rodent exposure was self-reported and included rodents in homes, rodent burrows around the house or killing rodents found in homes. When evaluating factors at baseline, HIV status was the only variable which remained significantly associated with seropositivity in the

Factor	N	n	OR crude [95% CI]	P crude	OR adjusted ^a [95% CI]	P adjusted ^a
Age (years)	240	119	1.02 [0.98-1.08]	0.294	_	_
GA at enrolment ^b (weeks)	234	116	0.98 [0.95-1.02]	0.259	_	_
Parity	240	119	1.15 [1.01-1.32]	0.040	1.13 [0.96-1.32]	0.128
Lives in a rural area						
No	110	62	1.66 [1.00-2.79]	0.050	1.42 [0.89-2.68]	0.126
Yes	130	57	Reference		Reference	
Educational level						
Post-secondary	167	77	Reference		Reference	
No Post-secondary	73	42	1.67 [0.95-2.98]	0.075	1.25 [0.66-2.40]	0.397
Occupation						
Student	15	8	Reference			
Housewife	39	20	1.20 [0.36-4.05]		_	_
Health professional	25	8	0.53 [0.14-2.01]	0.651 ^c	_	_
Informal Sector	100	54	1.37 [0.45-4.18]		_	_
Formal Sector	61	29	1.09 [0.35-3.49]		_	_
Knowledge of LF						
Good	30	11	Reference		Reference	
Poor	210	108	1.83 [0.84-4.17]	0.132	1.87 [0.81-4.48]	0.146
Exposure to rodents						
No	49	17	Reference		Reference	
Yes	191	102	2.14 [1.13-4.20]	0.022	2.32 [1.18-4.70]	0.015
Possible exposure to LF patients						
Unlikely	226	114	Reference			
Likely	14	5	0.53 [0.16-1.58]	0.269		
Fever during pregnancy				0 200		
No	198	94	Reference			
Yes	42	25	1.60 [0.83-3.20]	0.174		
Positive history of Lassa fever	12	25		0171		
No	235	116	Reference			
Yes	5	3	1.52 [0.25-11.7]	0.651		
Malaria during pregnancy				0.001		
No	143	65	Reference		Reference	
Vec	97	54	1.52 [0.90_2.55]	0.118	1.60 [0.91_2.85]	0.103
Hypertension		51	102[000 200]	0 110	100[091200]	0 100
No	235	116	Reference			
Vec	5	3	1.46 [0.24_11.3]	0.680		
Pregnancy_induced hypertension			140[024 113]	0.000		
No	232	114	Reference			
Vas	232	5	1.76 [0.42 8.76]	0.447		
Diabetes Mellitus ^d	0		1.10 [0.42-0.10]	0.44/		
No	221	117	Pafaranco		Pafaranco	
Vac	0	2		0.110		0.104
HIV infacted	7	Z	0.20 [0.04-1.19]	0.112	0.30 [0.02-1.01]	0.104
No.	224	107	Defenen		Defence	
INU	220	10/	Reference		Kelerence	

Table 3. Factors associated with Lassa maternal IgG seropositivity at baseline or delivery.

(Continued)

Table 3. (Continued)

Factor	N	n	OR crude [95% CI]	P crude	OR adjusted ^a [95% CI]	P adjusted ^a
Yes	14	12	7.01 [1.84-45.9]	0.012	5.16 [1.22-36.4]	0.046

Note: For statistical significance, Bonferroni p <0.005; CI- confidence interval; GA- gestational age; LF- Lassa fever; n- number of seropositive women; N- total number of women who were enrolled into the study (240); OR- odds ratio.

^a Adjusted for factors which were marginally significant in the univariate regression (crude p<0.25), except collinear terms.

^b Data missing for 6 women.

^c Overall P value.

^d None of the women with gestational diabetes were seropositive thus diabetes is evaluated as one group to prevent separation in the model.

https://doi.org/10.1371/journal.pntd.0011354.t003

multivariable regression, and when we accounted for multiple comparisons, none of the variables were significantly associated with baseline seropositivity. <u>S1 Table</u> summarises the risk factors for seropositivity at baseline. At baseline, two pregnant women had equivocal results and were excluded from the baseline seroprevalence analysis. There were no equivocal results at delivery. Of the two patients who had equivocal results at baseline, we found that at delivery one was positive, and the other was negative.

Lassa IgG seroconversion and seroreversion

Of the 240 participants who were recruited, 67 (27.9%) were lost-to-follow-up (LTFU). The difference between various maternal characteristics amongst those LTFU and those retained was not statistically significant for most variables. The only maternal characteristic affected significantly by LTFU was the educational level, where 43.6% of women with secondary education were LTFU compared to 27.3% with primary education and 22.2% with post-secondary education (p = 0.007), (S2 Table). This resulted in those with post-secondary education being over-represented in this cohort of pregnant women.

One hundred and seventy-three pregnant women participated at both baseline and delivery, of these, 106 were seronegative at baseline (Lassa naïve/susceptible). Amongst those who were susceptible at baseline, 22 satisfied the criteria for seroconversion, with an estimated seroconversion risk of 20.8% [95% CI: 14.1–29.4%] amongst pregnant women. Amongst the women who seroconverted, the median time spent in the study was 144 days [IQR: 125.5–183.0 days].

Overall, the mean age for LASV IgG seroconversion was 32.9 ± 5.9 years [range: 21-41] but was not significantly different from those who remained at risk (Mann-Whitney U p = 0.099). At baseline, five women had self-reported history of Lassa fever, three of these women were seronegative at baseline and had samples at both baseline and delivery. Of these, one satisfied the criteria for seroconversion. There was a positive history of fever during pregnancy in six of the women who seroconverted, Table 4.

In a multivariable regression model (Table 4), a positive history of rodent exposure was associated with increased odds of seroconversion (OR, 11.77 [1.46–1523], p = 0.014). However, when accounting for multiple comparisons with the Bonferroni correction, none of the variables remained statistically significant.

Seroreversion was not one of the initial objectives of this study. However, of the 67 pregnant women who were seropositive at baseline and who returned at labour/delivery, nine serore-verted (13.4% [95% CI: 7.2–23.6%]), that is had a LASV IgG index value of 0.9 or below (\leq 0.9) at labour/delivery and this decrease was over four-fold for all nine sample pairs. Amongst the women who seroreverted, the median time spent in the study was 163 days [IQR: 124.8–182.8 days].

Table 4.	Factors associated	with Lassa	IgG seroconversion	during pregnancy.
----------	--------------------	------------	--------------------	-------------------

Factor	N	n	χ ² P value ^a	OR crude [95% CI]	P crude	OR adjusted ^b [95% CI]	P adjusted ^b
Total	106	22					
Age (years)	106	22	0.099	1.06 [0.97–1.17]	0.211	1.03 [0.94–1.14]	0.557
Parity	106	22	0.318	1.20 [0.90-1.59]	0.204		_
Lives in a rural area							
No	42	12	0.142	2.13 [0.84-5.50]	0.112	2.34 [0.86-6.77]	0.097
Yes	64	10		Reference		Reference	
Educational level							
Post-secondary	83	14		Reference		Reference	
No post-secondary	23	8	0.081	2.63 [0.94–7.19]	0.066	2.65 [0.82-8.49]	0.101
Occupation							
Student	8	3		Reference			_
Housewife	15	2		0.29 [0.04–1.92]			_
НСР	17	4	0.713	0.52 [0.09-3.04]	0.724 ^c	_	_
Informal Sector	34	6		0.36 [0.07-1.88]			_
Formal Sector	32	7		0.46 [0.09-2.39]			_
Knowledge of LF risk & transmission							
Good	17	2		Reference		_	_
Poor	89	20	0.515	1.83 [0.51-9.79]	0.381	_	_
Exposure to rodents							
No	22	0		Reference		Reference	
Yes	83	22	0.006	16.19 [2.07-2088]	0.003	11.77 [1.46–1523]	0.014
Possible exposure to LF patients							
Unlikely	97	20		Reference		—	—
Likely	9	2	1.00	1.26 [0.22-5.18]	0.768	—	—
Fever during pregnancy							
No	87	16		Reference		Reference	
Yes	19	6	0.218	2.08 [0.68-6.02]	0.192	1.80 [0.48-6.26]	0.368
History of Lassa fever							
No	103	21		Reference			_
Yes	3	1	0.506	2.30[0.20-18.24]	0.451		_
Malaria during pregnancy							
No	67	13		Reference			_
Yes	39	9	0.804	1.25 [0.48-3.21]	0.635		_
Hypertension							
No	104	21		Reference			_
Yes	2	1	0.881	3.88 [0.30-49.7]	0.265		_
Pregnancy induced hypertension							
No	103	20		Reference		Reference	
Yes	3	2	0.205	6.79 [0.86–77.2]	0.069	8.18 [0.95-101]	0.055
Diabetes mellitus							
No	104	22		Reference			_
Yes	2	0	1.00	0.73 [0.005-9.44]	0.838		
Gestational Diabetes							
No	102	22		Reference			
Yes	4	0	0.578	0.39 [0.003-3.95]	0.493		
HIV-infected							

(Continued)

Table 4. (Continued)

Factor	N	n	χ ² P value ^a	OR crude [95% CI]	P crude	OR adjusted ^b [95% CI]	P adjusted ^b
No	105	21		Reference		Reference	
Yes	1	1	0.207	11.79 [0.61–1745]	0.101	4.03 [0.17-654]	0.397

Note: For statistical significance, Bonferroni p<0.008; CI- confidence interval; GA- gestational age; HCP- health professionals; LF- Lassa fever; n- number of women who seroconverted; N- total number of women who were seronegative at baseline (enrolment); OR- odds ratio; χ^2 - Chi squared.

^a Fisher's Exact p value or χ^2 p value as appropriate.

^b Adjusted for factors which were marginally significant in the univariate regression (crude p<0.25), except collinear terms.

^c Overall P value.

https://doi.org/10.1371/journal.pntd.0011354.t004

Discussion

The study shows a high prevalence of prior LASV IgG antibodies (49.6%) in pregnant women attending antenatal clinics in this hyperendemic region (Central Edo State, Nigeria). This suggests a susceptibility of 50% in pregnant women during the study period, which is higher than susceptibility reported in the general population in this hyperendemic region (42%). [19] The recorded increase in prevalence seen between enrolment (40%) and delivery (45.7%) may be explained by the fact that 60–85% of the delivery data from our cohort was recorded after the yearly Lassa peak period in Nigeria. [3, 28] Over the last couple of years, data from ISTH and the region suggest an increasing trend in occurrence of Lassa fever. [3, 55] For instance, in 2018 the total number of cases was 218 and increased to 289 in 2019 and 386 in 2020, compared to 112 in 2017 which suggest that our data was collected in a year where there was an increase in Lassa incidence. [56] Our findings therefore need to be taken in light of this consideration.

Our study had two time points, as such, we could not estimate the exact time at which seroconversion occurred or identify the time at which women were lost-to-follow-up. As a result, we could not estimate the incidence rate but rather report on the incidence risk (seroconversion risk). The seroconversion risk in this cohort was 20.8%, which is much higher than seroconversion observed in the general population in the non-endemic region of Mali (4–8%)[24] but similar to that reported in other hyperendemic areas such as in Sierra Leone where seroconversion was reported to be between 5–22%. [6] It is unlikely that all of these are new incidences of Lassa fever, and supports the view that reinfection may occur and results in a mild form of illness. [6] Nevertheless, the findings underscore the need for larger prospective studies to understand the clinical course of LASV infection in pregnancy and the need for preventative and therapeutic solutions tailored to pregnant women.

Our data suggest a substantial seroreversion risk during the study period (13.4%), which is much higher than that estimated in the general population in Sierra Leone (6.4%). [6]. The presence of seroreversion has an impact on the estimation of the burden of Lassa fever in this region, particularly, the estimation of susceptibility in pregnancy and suggests that the prevalence obtained in this, and other cohorts are an underestimate of the actual proportion of women of childbearing age who present at pregnancy with prior LASV exposure. This is relevant because the impact of a vaccination programme depends in part on the risk of exposure, the susceptibility of the exposed population, and the probability of adverse outcomes once infected. Further studies are needed to provide a better understanding of the risk of seroreversion and the susceptibility of seroreverted pregnant women to LASV reinfection during pregnancy. Additionally, the occurrence of both seroconversion and seroreversion in this cohort suggests that these parameters would need to be considered for the development of Lassa vaccine efficacy, effectiveness, and utility models.

An alternative explanation for the observed seroconversion and seroreversion risk is potentially the cross-reactivity of the assay with other antigens, [57] which may result in false negatives or false positives. Similarly, lipaemia in the samples (Price C & Newman D 1997 ref. in Tate J et al. [57]), as well as repeated freeze-thaw cycles, [58] may result in false positives or false negatives and support the need for further studies. An IgG neutralisation test may have facilitated confirmation of seroconversion and seroreversion. [59] and would be worth exploring in another study.

The relatively high effect sizes for the association of rodent exposure to seroprevalence [OR, 2.32] and seroconversion [OR, 11.77], suggests that similar to the general population, pregnant women were more likely to get Lassa infection from zoonotic spill over rather than from human exposures. [42, 43] With an attributable proportion of 35.0%, the data suggest that infections might be preventable by avoiding rodent exposure and conditions which facilitate infestation and the risk of human-rodent contact. Ecological studies from Nigeria suggest that while in most areas the main reservoir for Lassa fever (*Mastomys natalensis*) was found indoors, in some urban areas in Edo state, these small mammals were mainly found outdoors. [60] As such, given that our study did not categorise the type of rodents or quantify exposure, future studies particularly One Health studies. [17, 61] are needed to provide a better understanding of the avenues of human-rodent interaction in pregnancy to support development of effective control strategies. [17, 62–64] In the interim, advocating public health measures for rodent control such as proper sanitation and hygiene, improved housing and health education could support Lassa fever control efforts. [64, 65].

Contrary to previous studies, [66, 67] we found there was no significant difference between seropositivity amongst pregnant women living in rural areas compared to those in urban areas. There was also no association found between LASV seropositivity and educational level, unlike in previous studies in the general population in Edo state.[68] Our findings can probably be explained by the fact that post-secondary educated participants were over-represented in our cohort. However, community-based studies in this region have also reported high levels of formal education, [69] signifying that this region may have higher literacy levels. Despite the over-representation of post-secondary educated participants in this cohort, knowledge on Lassa fever transmission and risk factors was generally poor (87.5%). Further studies would be useful to understand the interrelation between education and seropositivity and identify knowledge gaps and behaviours that impede or promote Lassa control particularly in pregnant women.

Our cohort had a mean age of 31.6±5 years which is higher than reports on the national average in Nigeria, which in 2018 was 26 years. [70] The relatively high age of women in our cohort, may be explained by the over-representation of post-secondary educated (55.8%) and multiparous women (71.9%) in our cohort. Additionally, our cohort was recruited from health facilities and studies suggest that in Nigeria, younger women (<25years) are less likely to attend or deliver in health facilities. [71] Demographic surveillance from Nigeria suggest that adults between 21–40 years were the age group most affected by Lassa fever in 2019, and may account for the high seroprevalence (49.6%) observed in our cohort. [72] Further community based studies with a pregnant cohort may provide a better understanding of the effect of age on Lassa seropositivity.

We recorded an association between HIV infection and LASV IgG seropositivity. While these associations were not statistically significant, the effect sizes were high and warrant further inquiry particularly given the negative consequences of these diseases in the pregnant woman and her foetus. [7] It is also possible that any correlation between Lassa fever and HIV may simply reflect their relationship to socioeconomic disparities, given that both Lassa fever [5, 68] and HIV [73] are considered diseases of poverty; although evidence in HIV is conflicting and context-specific.

Limitations

The use of a cohort design brings the added risk of loss to follow up (LTFU); we addressed this by adjusting our sample size to accommodate LTFU at rates equivalent to the average differences generally observed between ANC and institutional delivery at ISTH. The actual LTFU was lower than we had accounted for in the sample size.

Most of the pregnant women enrolled in the study were in their second trimester of pregnancy. This is unsurprising because pregnant women in Sub-Saharan Africa tend to attend ANC after the first trimester. [74–77] However, it did not affect the estimate for seroconversion risk because the definition for seroconversion included a time frame allowing only women with over 30 days in the study to be evaluated as seroconverts.

While a seroconversion risk of 20.8% is relatively high, in terms of absolute numbers, the number of seroconversion events was small and given the extremely wide confidence intervals, the estimates for potential risk factors may not be precise, however, the study highlights potential areas for further inquiry and provides a baseline on seroconversion in pregnancy which can facilitate design of larger prospective studies.

Conclusion

The study found a prevalence of 49.6% for Lassa IgG antibodies in pregnant women in Edo State, Nigeria with an estimated attributable risk proportion of 35% for rodent exposure. However, the evidence on rodent exposure in our study was subjective and further studies are needed to provide a better understanding of the avenues of human-rodent interaction and Lassa control in pregnancy. In the interim, rodent control measures would be beneficial. While the evidence is not conclusive, our study provides a baseline for seroconversion in pregnancy to facilitate future prospective studies. Moreover, the estimated seroconversion risk of 20.8% is similar to that observed in the general population in hyperendemic areas, and while most of these seroconversions may not be new infections, given the high risk of adverse outcomes in pregnancy, our study supports the need for preventative and therapeutic options against Lassa fever in pregnancy. The occurrence of seroreversion suggests that the prevalence obtained in this, and other cohorts may be an underestimate of the actual proportion of women of childbearing age who present at pregnancy with prior Lassa exposure. Additionally, the occurrence of both seroconversion and seroreversion in this cohort suggests that these parameters would need to be considered for the development of Lassa vaccine efficacy, effectiveness, and utility models. This study provides valuable information for estimation of the epidemiologic disease burden of Lassa fever, which contributes to an estimation of disease impact in the short term and long term, and as such, allows for planning and allocation of resources, and facilitates monitoring and evaluation of disease control policies.

Supporting information

S1 File. Sample size calculation and attributable risk proportion. (PDF)
S2 File. Guidance for obtaining informed consent. (PDF)
S3 File. Enrolment or baseline questionnaire. (PDF)
S4 File. Delivery questionnaire. (PDF)

S1 Table. Factors associated with Lassa maternal IgG seropositivity at baseline. (DOCX)

S2 Table. Characteristics of participants lost-to-follow-up (LTFU) & an estimation of the effects of LTFU. (DOCX)

Acknowledgments

We would like to thank all of the members of the obstetrics and gynaecology department particularly Dr GBO Okome, Prof. FO Okogbo, Dr FI Omorogbe, Dr. P Isabu, Dr. A Ajekweneh, Dr E Ehidiamen, Ekosuehi Francis, Eigbefoh Francis, Yaya Oluwaseun, Oladeinde Oladapo, Ogbiti Mark, Omoregbee Hilda, Oboh Samson, Dantani Danladi, Esele Ekuaze, Asogwa Louis, Lawal Qudus, Oshodin Akhere, Ogunbor Philomena, Nwokoro Augustina and Okokhere Juliana for their support throughout the project. We also extend our gratitude to other members of Irrua Specialist Teaching Hospital especially Dr Ephraim Ogbaeni, Dr Cyril Erameh and Dr Peter Akhideno for their support throughout the project.

Author Contributions

Conceptualization: Nzelle Delphine Kayem, Proochista Ariana, Peter Horby.

- **Data curation:** Nzelle Delphine Kayem, Joseph Okoeguale, Mojeed Momoh, Antonia Njoku, Reuben Eifediyi, Xavier Enodiana, Hilary Ngwu, Wilfred Irhiogbe, George Odigie, Lyndsey Castle.
- Formal analysis: Nzelle Delphine Kayem.

Funding acquisition: Nzelle Delphine Kayem, Peter Horby.

Investigation: Nzelle Delphine Kayem, Yemisi Ighodalo, Thomas Olokor, Sophie Duraffour, Lisa Oestereich, Stephan Gunther, Peter Horby.

Methodology: Nzelle Delphine Kayem, Proochista Ariana, Stephan Gunther, Peter Horby.

Project administration: Nzelle Delphine Kayem, Sylvanus Okogbenin, Joseph Okoeguale.

Resources: Nzelle Delphine Kayem, Sylvanus Okogbenin, Peter Horby.

Software: Nzelle Delphine Kayem.

Supervision: Proochista Ariana, Peter Horby.

Validation: Nzelle Delphine Kayem, Prabin Dahal, Peter Horby.

Visualization: Nzelle Delphine Kayem.

Writing - original draft: Nzelle Delphine Kayem.

Writing – review & editing: Nzelle Delphine Kayem, Sylvanus Okogbenin, Joseph Okoeguale, Mojeed Momoh, Antonia Njoku, Reuben Eifediyi, Xavier Enodiana, Hilary Ngwu, Wilfred Irhiogbe, Yemisi Ighodalo, Thomas Olokor, George Odigie, Lyndsey Castle, Sophie Duraffour, Lisa Oestereich, Prabin Dahal, Proochista Ariana, Stephan Gunther, Peter Horby.

References

- Shaffer JG, Grant DS, Schieffelin JS, Boisen ML, Goba A, Hartnett JN, et al. Lassa Fever in Post-Conflict Sierra Leone. PLoS Negl Trop Dis. 2014; 8(3):e2748. https://doi.org/10.1371/journal.pntd.0002748 PMID: 24651047
- Bausch DG, Demby AH, Coulibaly M, Kanu J, Goba A, Bah A, et al. Lassa Fever in Guinea: I. Epidemiology of Human Disease and Clinical Observations. Vector Borne Zoonotic Dis. 2001; 1(4):269–81. https://doi.org/10.1089/15303660160025903 PMID: 12653127
- Akpede GO, Asogun DA, Okogbenin SA, Dawodu SO, Momoh MO, Dongo AE, et al. Caseload and Case Fatality of Lassa Fever in Nigeria, 2001–2018: A Specialist Center's Experience and Its Implications. Front Public Health. 2019; 7:170. PubMed Central PMCID: PMC6603170. https://doi.org/10. 3389/fpubh.2019.00170 PMID: 31294014
- Mylne AQ, Pigott DM, Longbottom J, Shearer F, Duda KA, Messina JP, et al. Mapping the zoonotic niche of Lassa fever in Africa. Trans R Soc Trop Med Hyg. 2015; 109(8):483–92. Epub 2015/06/19. https://doi.org/10.1093/trstmh/trv047 PMID: 26085474; PubMed Central PMCID: PMC4501400.
- Richmond KJ, Baglole DJ. Lassa fever: epidemiology, clinical features, and social consequences. BMJ 2003; 327(7426):1271–5. https://doi.org/10.1136/bmj.327.7426.1271 PMID: 14644972
- McCormick JB, Webb PA, Krebs JW, Johnson KM, Smith ES. A prospective study of the epidemiology and ecology of Lassa fever. J Infect Dis. 1987; 155(3):437–44. https://doi.org/10.1093/infdis/155.3.437 PMID: 3805771
- Kayem ND, Benson C, Aye CYL, Barker S, Tome M, Kennedy S, et al. Lassa fever in pregnancy: a systematic review and meta-analysis. Trans R Soc Trop Med Hyg. 2020; 114(5):385–96. Epub 2020/03/ 04. https://doi.org/10.1093/trstmh/traa011 PMID: 32125412.
- Price ME, Fisher-Hoch SP, Craven RB, McCormick JB. A prospective study of maternal and fetal outcome in acute Lassa fever infection during pregnancy. BMJ. 1988; 297(6648):584–7. <u>https://doi.org/10.1136/bmj.297.6648.584</u> PMID: 3139220
- Okogbenin S, Okoeguale J, Akpede G, Colubri A, Barnes KG, Mehta S, et al. Retrospective Cohort Study of Lassa Fever in Pregnancy, Southern Nigeria. Emerging Infect Dis. 2019; 25(8). Epub 2019/07/ 17. https://doi.org/10.3201/eid2508.181299 PMID: 31310586; PubMed Central PMCID: PMC6649346.
- Ippolito G, Feldmann H, Lanini S, Vairo F, Di Caro A, Capobianchi MR, et al. Viral hemorrhagic fevers: advancing the level of treatment. BMC Medicine. 2012; 10(1):31. <u>https://doi.org/10.1186/1741-7015-10-31 PMID: 22458265</u>
- Eberhardt KA, Mischlinger J, Jordan S, Groger M, Günther S, Ramharter M. Ribavirin for the treatment of Lassa fever: A systematic review and meta-analysis. International Journal of Infectious Diseases. 2019; 87:15–20. https://doi.org/10.1016/j.ijjd.2019.07.015 PMID: 31357056
- 12. Heyrana K, Byers HM, Stratton P. Increasing the Participation of Pregnant Women in Clinical Trials. JAMA. 2018; 320(20):2077–8. https://doi.org/10.1001/jama.2018.17716 PMID: 30422300
- 13. The Coalition for Epidemic Preparedness Innovations (CEPI). Lassa vaccine portfolio 2022 [cited 2023 16 January]. Available from: https://cepi.net/research_dev/our-portfolio/.
- 14. Inovio Pharmaceuticals Inc. INOVIO Announces First Subject Dosed in Phase 1B Clinical Trial for its DNA Vaccine Against Lassa Fever, INO-4500, in West Africa 2020 [cited 2020 23/02]. Available from: https://www.prnewswire.com/news-releases/inovio-announces-first-subject-dosed-in-phase-1b-clinical-trial-for-its-dna-vaccine-against-lassa-fever-ino-4500-in-west-africa-301233010.html.
- 15. The Coalition for Epidemic Preparedness (CEPI). As CEPI ramps up vaccine research, a new study warns Lassa fever is set to expand its reach across Africa 2023 [cited 2023 16 January]. Available from: https://cepi.net/news_cepi/as-cepi-ramps-up-vaccine-research-a-new-study-warns-lassa-fever-is-set-to-expand-its-reach-across-africa/.
- 16. Kayem ND. Viral haemorrhagic fevers in pregnancy with a focus on lassa fever [DPhil]. Oxford,UK: University of Oxford; 2021.
- Garry RF. Lassa fever—the road ahead. Nature Reviews Microbiology. 2022. https://doi.org/10.1038/ s41579-022-00789-8 PMID: 36097163
- Cutts Felicity T, Hanson M. Seroepidemiology: an underused tool for designing and monitoring vaccination programmes in low- and middle-income countries. Tropical Medicine & International Health. 2016; 21(9):1086–98. https://doi.org/10.1111/tmi.12737 PMID: 27300255
- Tobin E, Asogun D, Akpede N, Adomeh D, Odia I, Gunther S. Lassa fever in Nigeria: Insights into seroprevalence and risk factors in rural Edo State: A pilot study. Journal of Medicine in the Tropics. 2015; 17 (2):51–5. https://doi.org/10.4103/2276-7096.162258
- Tomori O, Fabiyi A, Sorungbe A, Smith A, McCormick JB. Viral hemorrhagic fever antibodies in Nigerian populations. Am J Trop Med Hyg. 1988; 38(2):407–10. Epub 1988/03/01. https://doi.org/10.4269/ajtmh. 1988.38.407 PMID: 3128130.

- Lukashevich IS, Clegg JCS, Sidibe K. Lassa virus activity in Guinea: Distribution of human antiviral antibody defined using enzyme-linked immunosorbent assay with recombinant antigen. Journal of Medical Virology. 1993; 40(3):210–7. https://doi.org/10.1002/jmv.1890400308 PMID: 8355019
- Yalley-Ogunro JE, Frame JD, Hanson AP. Endemic Lassa fever in Liberia. VI. Village serological surveys for evidence of Lassa virus activity in Lofa County, Liberia. Transactions of The Royal Society of Tropical Medicine and Hygiene. 1984; 78(6):764–70. <u>https://doi.org/10.1016/0035-9203(84)90013-0</u> PMID: 6398530
- Bloch A. A serological survey of Lassa fever in Liberia. Bulletin of the World Health Organization. 1978; 56(5):811–3. PMC2395672. PMID: 310723
- Safronetz D, Sogoba N, Diawara SI, Bane S, Rosenke K, Maiga O, et al. Annual Incidence of Lassa Virus Infection in Southern Mali. Am J Trop Med Hyg. 2017; 96(4):944–6. PMC5392646. <u>https://doi.org/ 10.4269/ajtmh.16-0821</u> PMID: 28093544
- Bausch DG, Rollin PE, Demby AH, Coulibaly M, Kanu J, Conteh AS, et al. Diagnosis and Clinical Virology of Lassa Fever as Evaluated by Enzyme-Linked Immunosorbent Assay, Indirect Fluorescent-Antibody Test, and Virus Isolation. J Clin Microbiol. 2000; 38(7):2670–7. PMC86994. https://doi.org/10. 1128/JCM.38.7.2670-2677.2000 PMID: 10878062
- Cauchemez S, Horby P, Fox A, Mai le Q, Thanh le T, Thai PQ, et al. Influenza infection rates, measurement errors and the interpretation of paired serology. PLoS pathogens. 2012; 8(12):e1003061. Epub 2012/12/29. https://doi.org/10.1371/journal.ppat.1003061 PMID: 23271967; PubMed Central PMCID: PMC3521724
- Ohuma EO, Okiro EA, Ochola R, Sande CJ, Cane PA, Medley GF, et al. The Natural History of Respiratory Syncytial Virus in a Birth Cohort: The Influence of Age and Previous Infection on Reinfection and Disease. 2012; 176(9):794–802. https://doi.org/10.1093/aje/kws257.
- Okoro OA, Bamgboye E, Dan-Nwafor C, Umeokonkwo C, Ilori E, Yashe R, et al. Descriptive epidemiology of Lassa fever in Nigeria, 2012–2017. Pan African Medical Journal [Internet]. 2020 5 October 2020 [cited 2020 5 October]; 37. Available from: https://www.panafrican-med-journal.com/content/article/37/ 15/full. https://doi.org/10.11604/pamj.2020.37.15.21160 PMID: 33062117
- Tey N-P, Lai S-I. Correlates of and Barriers to the Utilization of Health Services for Delivery in South Asia and Sub-Saharan Africa. TheScientificWorldJournal. 2013; 2013:11. https://doi.org/10.1155/2013/ 423403.
- Stekelenburg J, Kyanamina S, Mukelabai M, Wolffers I, Roosmalen J. Waiting too long: low use of maternal health services in Kalabo, Zambia. Tropical Medicine & International Health. 2004; 9(3):390– 8. https://doi.org/10.1111/j.1365-3156.2004.01202.x PMID: 14996369
- Harris PA, Taylor R, Minor BL, Elliott V, Fernandez M, O'Neal L, et al. The REDCap consortium: Building an international community of software platform partners. J Biomed Inform. 2019; 95:103208. https://doi.org/10.1016/j.jbi.2019.103208 PMID: 31078660
- Ballard JL, Khoury JC, Wedig K, Wang L, Eilers-Walsman BL, Lipp R. New Ballard Score, expanded to include extremely premature infants. J Pediatr. 1991; 119(3):417–23. https://doi.org/10.1016/s0022-3476(05)82056-6 PMID: 1880657
- Raabe V, Koehler J. Laboratory Diagnosis of Lassa Fever. Journal of Clinical Microbiology. 2017; 55 (6):1629–37. https://doi.org/10.1128/JCM.00170-17 PMID: 28404674
- Branco LM, Grove JN, Boisen ML, Shaffer JG, Goba A, Fullah M, et al. Emerging trends in Lassa fever: redefining the role of immunoglobulin M and inflammation in diagnosing acute infection. Virology Journal. 2011; 8:478–. PMC3223505. https://doi.org/10.1186/1743-422X-8-478 PMID: 22023795
- **35.** Gabriel M, Adomeh DI, Ehimuan J, Oyakhilome J, Omomoh EO, Ighodalo Y, et al. Development and evaluation of antibody-capture immunoassays for detection of Lassa virus nucleoprotein-specific immunoglobulin M and G. PLoS Negl Trop Dis. 2018; 12(3):e0006361. https://doi.org/10.1371/journal.pntd. 0006361 PMID: 29596412
- Johnson KM, McCormick JB, Webb PA, Smith ES, Elliott LH, King IJ. Clinical virology of Lassa fever in hospitalized patients. The Journal of infectious diseases. 1987; 155(3):456–64. Epub 1987/03/01. https://doi.org/10.1093/infdis/155.3.456 PMID: 3805773.
- Takah NF, Brangel P, Shrestha P, Peeling R. Sensitivity and specificity of diagnostic tests for Lassa fever: a systematic review. BMC Infectious Diseases. 2019; 19(1):647. https://doi.org/10.1186/s12879-019-4242-6.
- Emperador DM, Yimer SA, Mazzola LT, Norheim G, Kelly-Cirino C. Diagnostic applications for Lassa fever in limited-resource settings. BMJ Global Health. 2019; 4(Suppl 2):e001119. <u>https://doi.org/10. 1136/bmjgh-2018-001119</u> PMID: 30899576
- **39.** R CoreTeam. R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing; 2020.

- Brown LD, Cai TT, DasGupta A. Interval Estimation for a Binomial Proportion. Statist Sci. 2001; 16 (2):101–33. https://doi.org/10.1214/ss/1009213286
- 41. Steyerberg EW. Clinical Prediction Models: A Practical Approach to Development, Validation, and Updating. Switzerland: Springer Nature; 2019 [cited 2020 18 Nov].
- World Health Organization. Introduction to Lassa fever: Managing infectious hazards. Geneva, Switzerland: 2018.
- Lo Iacono G, Cunningham AA, Fichet-Calvet E, Garry RF, Grant DS, Khan SH, et al. Using Modelling to Disentangle the Relative Contributions of Zoonotic and Anthroponotic Transmission: The Case of Lassa Fever. PLoS Negl Trop Dis. 2015; 9(1):e3398. https://doi.org/10.1371/journal.pntd.0003398
 PMID: 25569707
- Firth D. Bias Reduction of Maximum Likelihood Estimates. Biometrika. 1993; 80(1):27–38. <u>https://doi.org/10.2307/2336755</u>.
- Šinkovec H, Geroldinger A, Heinze G. Bring More Data!—A Good Advice? Removing Separation in Logistic Regression by Increasing Sample Size. Int J Environ Res Public Health. 2019; 16(23). <u>https:// doi.org/10.3390/ijerph16234658</u> PMID: 31766753
- Chowdhury MZI, Turin TC. Variable selection strategies and its importance in clinical prediction modelling. Family Medicine and Community Health. 2020; 8(1):e000262. <u>https://doi.org/10.1136/fmch-2019-000262</u> PMID: 32148735
- **47.** Armstrong RA. When to use the Bonferroni correction. Ophthalmic Physiol Opt. 2014; 34(5):502–8. Epub 20140402. https://doi.org/10.1111/opo.12131 PMID: 24697967.
- Altman DG, Royston P. The cost of dichotomising continuous variables. Bmj. 2006; 332(7549):1080. https://doi.org/10.1136/bmj.332.7549.1080 PMID: 16675816; PubMed Central PMCID: PMC1458573.
- Austin PC, Brunner LJ. Inflation of the type I error rate when a continuous confounding variable is categorized in logistic regression analyses. Stat Med. 2004; 23(7):1159–78. https://doi.org/10.1002/sim. 1687 PMID: 15057884.
- Okoko JB, Wesumperuma HL, Hart CA. The influence of prematurity and low birthweight on transplacental antibody transfer in a rural West African population. Tropical Medicine & International Health. 2001; 6(7):529–34. https://doi.org/10.1046/j.1365-3156.2001.00741.x.
- 51. van den Berg JP, Westerbeek EAM, Berbers GAM, van Gageldonk PGM, van der Klis FRM, van Elburg RM. Transplacental Transport of IgG Antibodies Specific for Pertussis, Diphtheria, Tetanus, Haemophilus influenzae Type b, and Neisseria meningitidis Serogroup C Is Lower in Preterm Compared With Term Infants. The Pediatric infectious disease journal. 2010; 29(9). https://doi.org/10.1097/inf. 0b013e3181dc4f77 PMID: 20803841
- 52. Ogolla S, Daud II, Asito AS, Sumba OP, Ouma C, Vulule J, et al. Reduced Transplacental Transfer of a Subset of Epstein-Barr Virus-Specific Antibodies to Neonates of Mothers Infected with Plasmodium falciparum Malaria during Pregnancy. Clinical and Vaccine Immunology: CVI. 2015; 22(11):1197–205. https://doi.org/10.1128/CVI.00270-15. PMC4622112.
- 53. Horby PW, Laurie KL, Cowling BJ, Engelhardt OG, Sturm-Ramirez K, Sanchez JL, et al. CONSISE statement on the reporting of Seroepidemiologic Studies for influenza (ROSES-I statement): an extension of the STROBE statement. Influenza Other Respir Viruses. 2017; 11(1):2–14. PMC5155648. https://doi.org/10.1111/irv.12411 PMID: 27417916
- 54. Taiwo I. Shapefiles zip: figshare; 2020. Available from: https://figshare.com/articles/online_resource/ Shapefiles_zip/11822316.
- 55. Asogun DA, Adomeh DI, Ehimuan J, Odia I, Hass M, Gabriel M, et al. Molecular Diagnostics for Lassa Fever at Irrua Specialist Teaching Hospital, Nigeria: Lessons Learnt from Two Years of Laboratory Operation. PLOS Neglected Tropical Diseases. 2012; 6(9):e1839. <u>https://doi.org/10.1371/journal.pntd.</u> 0001839 PMID: 23029594
- 56. Yaro CA, Kogi E, Opara KN, Batiha GE-S, Baty RS, Albrakati A, et al. Infection pattern, case fatality rate and spread of Lassa virus in Nigeria. BMC Infectious Diseases. 2021; 21(1):149. <u>https://doi.org/10. 1186/s12879-021-05837-x PMID: 33546623</u>
- 57. Tate J, Ward G. Interferences in immunoassay. Clin Biochem Rev. 2004; 25(2):105–20. PMID: 18458713.
- Fipps DR, Damato JJ, Brandt B, Burke DS. Effects of multiple freeze thaws and various temperatures on the reactivity of human immunodeficiency virus antibody using three detection assays. J Virol Methods. 1988; 20(2):127–32. https://doi.org/10.1016/0166-0934(88)90146-2 PMID: 3166465
- Olitsky PK, Casals J. Neutralization tests for diagnosis of human virus encephalitides. JAMA. 1947; 134 (15):1224–8. https://doi.org/10.1001/jama.1947.02880320014004 PMID: 20253933

- Olayemi A, Obadare A, Oyeyiola A, Fasogbon S, Igbokwe J, Igbahenah F, et al. Small mammal diversity and dynamics within Nigeria, with emphasis on reservoirs of the lassa virus. Systematics and Biodiversity. 2018; 16(2):118–27. https://doi.org/10.1080/14772000.2017.1358220
- Arruda LB, Haider N, Olayemi A, Simons D, Ehichioya D, Yinka-Ogunleye A, et al. The niche of One Health approaches in Lassa fever surveillance and control. Annals of Clinical Microbiology and Antimicrobials. 2021; 20(1):29. https://doi.org/10.1186/s12941-021-00431-0 PMID: 33894784
- Mari Saez A, Cherif Haidara M, Camara A, Kourouma F, Sage M, Magassouba NF, et al. Rodent control to fight Lassa fever: Evaluation and lessons learned from a 4-year study in Upper Guinea. PLOS Neglected Tropical Diseases. 2018; 12(11):e0006829. <u>https://doi.org/10.1371/journal.pntd.0006829</u> PMID: 30399142
- Mariën J, Borremans B, Kourouma F, Baforday J, Rieger T, Günther S, et al. Evaluation of rodent control to fight Lassa fever based on field data and mathematical modelling. Emerging Microbes & Infections. 2019; 8(1):640–9. https://doi.org/10.1080/22221751.2019.1605846 PMID: 31007139
- Clark J, Yakob L, Douno M, Lamine J, Magassouba NF, Fichet-Calvet E, et al. Domestic risk factors for increased rodent abundance in a Lassa fever endemic region of rural Upper Guinea. Scientific Reports. 2021; 11(1):20698. https://doi.org/10.1038/s41598-021-00113-z PMID: 34667210
- Abdullahi IN, Anka AU, Ghamba PE, Onukegbe NB, Amadu DO, Salami MO. Need for preventive and control measures for Lassa fever through the One Health strategic approach. Proceedings of Singapore Healthcare. 2020; 29(3):190–4. https://doi.org/10.1177/2010105820932616
- 66. Hallam HJ, Hallam S, Rodriguez SE, Barrett ADT, Beasley DWC, Chua A, et al. Baseline mapping of Lassa fever virology, epidemiology and vaccine research and development. npj Vaccines. 2018; 3 (1):11. https://doi.org/10.1038/s41541-018-0049-5 PMID: 29581897
- 67. Kernéis S, Koivogui L, Magassouba NF, Koulemou K, Lewis R, Aplogan A, et al. Prevalence and Risk Factors of Lassa Seropositivity in Inhabitants of the Forest Region of Guinea: A Cross-Sectional Study. PLoS Negl Trop Dis. 2009; 3(11):e548. PMC2771900. https://doi.org/10.1371/journal.pntd.0000548 PMID: 19924222
- Tobin EA, Asogun D, Happi C, Ogbaini E, Gunther S. Risk factors for Lassa fever in endemic communities of Edo State, Nigeria. International Journal of Infectious Diseases. 2014; 21:258–9. https://doi.org/ 10.1016/j.ijid.2014.03.958.
- 69. Ekechi HU, Ibeneme C, Ogunniyi B, Awosanya E, Gbadebo B, Usman A, et al. Factors associated with a confirmed Lassa fever outbreak in Eguare community of Esan West, Edo State, Nigeria: January-March, 2019. Journal of Interventional Epidemiology and Public Health [Internet]. 2020 5 October 2020 [cited 2020 5 October]; 3. Available from: https://www.afenet-journal.net/content/article/3/1/full.
- Ayotunde T, Mary O, Melvin AO, Faniyi FF. Maternal age at birth and under-5 mortality in Nigeria. East Afr J Public Health. 2009; 6(1):11–4. https://doi.org/10.4314/eajph.v6i1.45735 PMID: 20000056.
- Adewuyi EO, Khanal V, Zhao Y, David L, Bamidele OD, Auta A. Home childbirth among young mothers aged 15–24 years in Nigeria: a national population-based cross-sectional study. BMJ Open. 2019; 9(9): e025494. https://doi.org/10.1136/bmjopen-2018-025494 PMID: 31537553
- 72. Grace J-UA, Egoh IJ, Udensi N. Epidemiological trends of Lassa fever in Nigeria from 2015–2021: A review. Ther Adv Infect Dis. 2021; 8:20499361211058252–. <u>https://doi.org/10.1177/</u>20499361211058252 PMID: 34868582.
- 73. Fox AM. The HIV-poverty thesis re-examined: poverty, wealth or inequality as a social determinant of HIV infection in sub-Saharan Africa? J Biosoc Sci. 2012; 44(4):459–80. Epub 2012/01/26. https://doi. org/10.1017/S0021932011000745 PMID: 22273351.
- 74. Gudayu TW, Woldeyohannes SM, Abdo AA. Timing and factors associated with first antenatal care booking among pregnant mothers in Gondar Town; North West Ethiopia. BMC Pregnancy and Childbirth. 2014; 14(1):287. https://doi.org/10.1186/1471-2393-14-287 PMID: 25154737
- 75. Kisuule I, Kaye DK, Najjuka F, Ssematimba SK, Arinda A, Nakitende G, et al. Timing and reasons for coming late for the first antenatal care visit by pregnant women at Mulago hospital, Kampala Uganda. BMC Pregnancy and Childbirth. 2013; 13(1):121. https://doi.org/10.1186/1471-2393-13-121 PMID: 23706142
- 76. Ndidi EP, Oseremen IG. Reasons Given by Pregnant Women for Late Initiation of Antenatal Care in the Niger Delta, Nigeria. Ghana Med J. 2010; 44(2):47–51. PMC2994152. PMID: 21327003
- 77. Tolefac PN, Halle-Ekane GE, Agbor VN, Sama CB, Ngwasiri C, Tebeu PM. Why do pregnant women present late for their first antenatal care consultation in Cameroon? Matern Health Neonatol Perinatol. 2017; 3(1):29. https://doi.org/10.1186/s40748-017-0067-8 PMID: 29255616